Author's response to reviews

Title: No red cell alloimmunization or change of clinical outcome after using fresh frozen cancellous allograft bone for acetabular reconstruction in revision hip arthroplasty: a follow up study

Authors:
Falk Mittag (falkmittag@med.uni-tuebingen.de)
Matthias Straub (matthias.straub@student.uni-tuebingen.de)
Richard Schäfer (richard.schaefer@med.uni-tuebingen.de)
Torsten Kluba (torsten.kluba@med.uni-tuebingen.de)
Ingmar Ipach (ingmar.ipach@med.uni-tuebingen.de)

Version: 2 Date: 11 May 2012

Author's response to reviews:

Dear editors,

we revised our manuscript as suggested (methods section: Research was carried out in compliance with the Helsinki Declaration, local ethics committee).

Sincerely yours,

Torsten Kluba et al.